

https://doi.org/10.1093/pnasnexus/pgaf175 Advance access publication 24 June 2025 Research Report

Improving predictability, reliability, and generalizability of brain-wide associations for cognitive abilities via multimodal stacking

Alina Tetereva (**D**^a, Annchen R. Knodt (**D**^b), Tracy R. Melzer (**D**^{c,d,e}, William van der Vliet (**D**^a, Bryn Gibson (**D**^a, Ahmad R. Hariri (**D**^b), Ethan T. Whitman^b, Jean Li (**D**^f, Farzane Lal Khakpoor (**D**^a, Jeremiah Deng (**D**^f, David Ireland (**D**^g, Sandhya Ramrakha (**D**^g) and Narun Pat (**D**^{a,*})

^aDepartment of Psychology, University of Otago, Dunedin 9016, New Zealand

^bDepartment of Psychology and Neuroscience, Duke University, Durham, NC 27710, USA

^cNew Zealand Brain Research Institute, Christchurch 8011, New Zealand

^dDepartment of Medicine, University of Otago, Christchurch 8011, New Zealand

^eSchool of Psychology, Speech and Hearing, University of Canterbury, Christchurch 8041, New Zealand

^fSchool of Computing, University of Otago, Dunedin 9016, New Zealand

^gDunedin Multidisciplinary Health and Development Research Unit, Department of Psychology, University of Otago, Dunedin 9016, New Zealand *To whom correspondence should be addressed: Email: narun.pat@otago.ac.nz

Edited By Stephen Fleming

Abstract

Brain-wide association studies (BWASs) have attempted to relate cognitive abilities with brain phenotypes, but have been challenged by issues such as predictability, test-retest reliability, and cross-cohort generalizability. To tackle these challenges, we proposed a machine learning "stacking" approach that draws information from whole-brain MRI across different modalities, from task-functional MRI (fMRI) contrasts and functional connectivity during tasks and rest to structural measures, into one prediction model. We benchmarked the benefits of stacking using the Human Connectome Projects: Young Adults (n = 873, 22–35 years old) and Human Connectome Projects —Aging (n = 504, 35–100 years old) and the Dunedin Multidisciplinary Health and Development Study (Dunedin Study, n = 754, 45 years old). For predictability, stacked models led to out-of-sample $r\sim0.5$ –0.6 when predicting cognitive abilities at the time of scanning, primarily driven by task-fMRI contrasts. Notably, using the Dunedin Study, we were able to predict participants' cognitive abilities at ages 7, 9, and 11 years using their multimodal MRI at age 45 years, with an out-of-sample r of 0.52. For test-retest reliability, stacked models reached an excellent level of reliability (interclass correlation > 0.75), even when we stacked only task-fMRI contrasts together. For generalizability, a stacked model with nontask MRI built from one dataset significantly predicted cognitive abilities in other datasets. Altogether, stacking is a viable approach to undertake the three challenges of BWAS for cognitive abilities.

Keywords: cognitive abilities, stacking, task fMRI, reliability, generalizability

Significance statement

Scientists have had limited success in predicting cognitive abilities from brain MRI. We proposed a machine learning method, called stacking, to draw information across different types of brain MRI. Using three large databases (n = 2,131, 22-100 years old), we found stacking to make the prediction of cognitive abilities (i) closer to actual cognitive scores when applied to a new individual, not part of the modeling process, (ii) reliable over times, and (iii) applicable to the data collected from different age groups and MRI scanners. Indeed, stacking, especially with fMRI task contrasts, allowed us to use MRI of people aged 45 years to predict their childhood cognitive abilities reasonably well. Accordingly, stacking may help MRI realize its potential to predict cognitive abilities.

Introduction

Individual differences in cognitive abilities are stable across the lifespan (1) and have relatively high heritability (2). They are key

indicators of educational achievements (3), career successes (4), well-being (5), socioeconomic stability (6), and health outcomes (7). Recent studies have also demonstrated a widespread



Competing Interest: The authors declare no competing interests. Received: November 28, 2024. Accepted: May 15, 2025

© The Author(s) 2025. Published by Oxford University Press on behalf of National Academy of Sciences. This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For commercial re-use, please contact reprints@oup.com for reprints and translation rights for reprints. All other permissions can be obtained through our RightsLink service via the Permissions link on the article page on our site—for further information please contact journals.permissions@oup.com. relationship between impairments in cognitive abilities and various psychopathological disorders (8, 9). Accordingly, relating individual differences in cognitive abilities to neuroimaging data has been a primary goal for cognitive neuroscientists, from both basic and applied science perspectives (10). This approach allows neuroscientists to scientifically quantify the presence of information related to cognitive abilities from each neuroimaging type or modality. It also paves the way for identifying neural indicators of cognitive abilities, which could be useful for understanding the etiology of neuro- and psychopathology (11). Indeed, a leading transdiagnostic framework for psychiatry, the Research Domain Criteria (RDoC), treats cognitive abilities as one of the main functional domains for psychopathology across diagnoses. Having a robust neural indicator of cognitive abilities, in addition to behavioral and genetic indicators, is central to the RDoC framework (12).

The availability of large-scale neuroimaging databases (13) and the accessibility of predictive modeling methodologies (11, 14) have provided encouraging avenues to pursue a neural indicator of cognitive abilities. Accordingly, several researchers have built prediction models to predict cognitive abilities from brain MRI signals and evaluated the models' performance on separate, unseen data in the so-called brain-wide association studies (BWASs) (11, 15). BWASs can be conducted using either univariate (also known as mass-univariate) or multivariate (also known as machine learning) methods to draw MRI information. While univariate methods draw data from one region/voxel at a time, multivariate methods draw MRI information across regions/voxels (11, 16, 17). These multivariate methods, from particular MRI modalities, appear to boost predictability for cognitive abilities (11, 16-18). For examples, Marek et al. (16) conducted BWAS on several large datasets and concluded that "More robust BWAS effects were detected for functional MRI (fMRI) (vs. structural), cognitive tests (vs. mental health questionnaires), and multivariate methods (vs. univariate)."

Akin to genome-wide association studies in genetics (19) that can integrate information across single nucleotide polymorphisms (SNPs) across the genome to create a predicted, propensity score for a phenotype of interest (e.g. cognitive abilities), known as polygenic scores, BWAS can also be used to create a similar predicted score for each individual based on his/her neuroimaging data. For instance, Marek et al. (16) used trained multivariate methods to predict cognitive abilities from brain MRI data using part of the data (known as training set) and applied the trained model to the unseen participants (known as test set). Participants in the test set then had a predicted score of their cognitive abilities, based on their brain MRI data. Yet, BWAS for cognitive abilities has faced several challenges, including but not limited to predictability, test-retest reliability, and generalizability, as detailed below (16, 20, 21). These challenges have led to headlines, such as "Cognitive Neuroscience at the Crossroads" (22) and "Scanning the Brain to Predict Behavior, a Daunting "Task" for MRI" (23). To address these issues, we (24, 25) have recently proposed a potential solution, "stacking" (26), which allows us to combine different modalities of MRI into one prediction model. In this study, we aim to formally benchmark the benefits of stacking in improving predictability, test-retest reliability, and generalizability, using three large-scale neuroimaging databases (27-29).

First, predictability, or out-of-sample prediction, pertains to the ability of prediction models to predict a target variable, e.g. cognitive abilities, based on features, e.g. fMRI data, of unseen participants, not part of the model-building processes (30). More specifically, we refer to an application of a validation within one dataset. Here, researchers usually take a relatively large dataset, split it into training and test sets, then build a model from the training set, and apply the model to the test set. In addition to doing one split, researchers could also apply a cross-validation (CV) strategy by splitting a dataset into different nonoverlapping training-test folds and looping through folds to calculate the average performance across the test sets (31, 32). Several earlier studies (33–35) did not apply any validation when predicting cognitive abilities from MRI, possibly causing inflated predictability (16). With proper data splitting, a recent meta-analysis (15) estimated the predictability of multivariate methods on brain MRI of different modalities with a validation for cognitive abilities to be a Pearson's correlation (r) of 0.42 on average.

While this level of predictability is encouraging, there is still room for improvement. Given that different MRI modalities may convey different information about the brain, drawing information across different MRI modalities could allow us to improve predictability further. Stacking enables researchers to draw information across MRI modalities, which seems to improve predictability over relying on any single MRI modality (24-26, 36, 37). In this framework (see Fig. 1), researchers first build "nonstacked" prediction models separately for each MRI set of features (e.g. cortical thickness or cortical area) and computed predicted values from each of these "nonstacked" models. They then treat these predicted values as features for "stacked" prediction models, allowing them to draw information across different MRI sets of features. Still, most studies use one single type of MRI to build prediction models. The popular choices include resting-state fMRI functional connectivity (FC) (Rest FC, or correlations in blood-oxygen-level-dependent [BOLD] time series across areas during rest) (38-40), task-fMRI FC (Task FC, or correlations in BOLD time series across brain regions during each task) (41-48), and structural MRI (including measures such as thickness, area and volume in cortical/subcortical areas) (49). While it is less common to use task-fMRI contrasts (Task Contrasts, or fMRI BOLD activity relevant to events in each task) to predict cognitive abilities, studies have started to show the superior predictability of contrasts from certain tasks, compared with other MRI modalities (24, 25, 39, 50). Nonetheless, previous attempts at stacking often ignored task contrasts (36, 37), and as a result, while improving over nonstacked models, they have not led to satisfactory predictive performance. We (24, 25) have started to show a boost in predictability when applying stacking to combine task contrasts with other MRI modalities. Here, to ensure the robustness of this approach, we examined the benefits of stacking task contrasts, along with other MRI modalities, on multiple large-scale datasets.

Second, test-retest reliability pertains to the rank stability of measurements across different time points, assuming the absence of significant changes between assessments (e.g. treatment exposure, injury, and/or disease progression) (48-52). For instance, if some people score higher than their peers at time one, they should also score higher than their peers at time two. To use prediction models as an indicator for individual differences in cognitive abilities, the predicted values should be reliable across time. A recent study challenged the test-retest reliability of task contrasts (20). Here, the researchers examined test-retest reliability of Task Contrasts in certain areas, known to be strongly elicited in each task, across two time points and found a poor level of test-retest reliability across different tasks and two datasets: the Human Connectome Project: Young Adults (HCP Young Adults) (29) and the Dunedin Multidisciplinary Health and Development Study (Dunedin Study) (28). This poor level of testretest reliability from Task Contrasts is concerning, especially when compared with the higher levels of test-retest reliability



Fig. 1. Overview of study methodology. We used three datasets: HCP Young Adults, HCP Aging, and Dunedin Multidisciplinary Health and Development Study (Dunedin Study). a) Machine Learning Pipeline. Here, we depict the process we used for building prediction models for testing predictability within each dataset. Briefly, we used nested cross-validation (CV) by splitting the data into outer folds with around 100 participants in each. In each outer-fold CV loop, we then treated one of the outer folds as an outer-fold test set and treated the rest as an outer-fold training set. We then divided each outer-fold training set into five inner folds and applied inner-fold CV to build prediction models in three steps. In the first step (known as a nonstacking layer), one of the inner folds was treated as an inner-fold validation set, and the rest was treated as an inner-fold training set in each inner-fold CV. We used grid search to tune prediction models for each set of features. In the second step (known as a stacking layer), we treated different combinations of the predicted values from separate sets of features to predict the cognitive abilities in separate "stacked" models. In the third step, we applied the already tuned models from the first and second steps to the outer-fold test set. b) Predictability. Here, we examined the predictive performance across outer-fold test sets within each dataset. c) Test–retest reliability. Here, we used HCP Young Adults and Dunedin Study and treated participants who were scanned twice across MRI sessions as the test set and the rest as the training set. We then examined the ICC of the predicted values in the test set between the first and second MRI sessions. d) Generalizability. Here, we examined the predictive performance of the models built from a different dataset. We treated one of the three datasets as a training set and the other two as two separate test sets. e) Age distribution. Here, we show the age of participants at the time of scanning in each dataset.

found from structural MRI, Rest FC, and Task FC (20, 53, 54). In fact, structural MRI provided reliability that was almost at the ceiling (20). Yet, these studies simply took task contrasts from certain

areas; they did not create prediction models or use multivariate methods and stacking to draw information across the whole brain and across different tasks/MRI modalities. It is possible that Task Contrasts could be more reliable once the models consider information across the whole brain, across different tasks, and across different MRI modalities. Following this conjecture, we (25) recently showed that multivariate methods and stacking substantively boosted reliability, reaching a much higher level of reliability in HCP Young Adults (29). To ensure the robustness of our findings, we need to test the benefits of this stacking approach in another, independent dataset: the Dunedin Study (28), for example.

Third, generalizability, or more specifically cross-cohort generalizability, pertains to the ability of prediction models built from one dataset to predict the cognitive abilities of participants of another dataset (21). Different datasets, for instance, use different MRI scanners, recruit participants from different cultures and age groups, or implement different cognitive ability measurements. Thus, while predictability within one dataset provides the performance of prediction models within specific, harmonized contexts of one dataset, generalizability across datasets allows us to gauge the performance of prediction models in broader contexts. This means that generalizability situates closer to how deployable the prediction models are in indicating cognitive abilities in the real world (21). Yet, only a few studies have investigated the generalizability of MRI prediction models for cognitive abilities, and most have focused on FC during rest and/or tasks (55–57). The generalizability of stacked models is currently unknown.

Our overarching goal is to benchmark the impact of stacking on predictability, test-retest reliability, and generalizability of MRI prediction models for cognitive abilities. To achieve this, we used three large-scale neuroimaging databases, including HCP Young Adults (29), Human Connectome Project-Aging (HCP Aging) (27), and Dunedin Study (28). The databases vary in various aspects, such as participants' age (see Fig. 1) and cultures, physical scanners, scanning parameters, and cognitive ability assessments. Note that, unlike our previous implementation of stacking (25), we also included Task FC in addition to Task Contrasts to capture wider information during task scanning. Specifically, here we built stacked models from eight different combinations of functional and structural MRI sets of features: "Task Contrast" including Task Contrasts from all of the tasks, "Task FC" including Task FC from all of the tasks, "Non Task" including Rest FC and structural MRI, "Task Contrast & FC" including task contrasts and task FC from all of the tasks, "All" including all sets of features, "All excluding Task Contrast" including every set of features except Task Contrasts, "All, excluding Task FC" including every set of features except Task FC, and "Resting and Task FC" including FC during rest and tasks.

For predictability (see Fig. 1), we applied nested CV within each dataset to evaluate the predictability of stacked models from multimodal MRI. To build the stacked models, we applied 16 combinations of multivariate predictive modeling algorithms (including Elastic Net (58), Support Vector Regression (SVR) (59), Random Forest (60), and XGBoost (61)). Moreover, the nature of the Dunedin Study's longitudinal measurements (28) allowed a unique opportunity for us to predict cognitive abilities from MRI data at the time of scanning (at the age of 45 years), but also at much earlier times (at the age of 7, 9, and 11 years), as well as to predict the residual scores that reflect relative changes in cognitive abilities during 36 years, compared with participants' peers (62). For test-retest reliability (see Fig. 1), we examined the rank stability of stacked models from participants who were scanned twice in HCP Young Adults and Dunedin Study. Lastly, for generalizability (see Fig. 1), we built stacked models from one dataset and evaluated their performance on the other two. Due to the

different tasks used in different datasets, we unfortunately could only examine the generalizability of the "Stacked: Non Task" model, which combined all MRI modalities that did not involve tasks.

Results

Predictability

We showed performance indices of stacked and nonstacked models for each predictive modeling algorithm and dataset in Figs. 2 and S1-S9 and their bootstrapped 95% CI in Figs. S10-S18. Overall, when predicting cognitive abilities at the time of scanning, the prediction models from multimodal MRI across different predictive modeling algorithms varied in their performance, reflected by Pearson's correlation (r) between predicted and observed values, ranging from around 0 to 0.6, across the three datasets. Notably, combining different sets of MRI features into stacked models constantly led to higher predictive performance. "Stacked: All," which included all sets of MRI features, gave rise to top-performing models across algorithms and the three datasets. Additionally, using Elastic Net across both nonstacking and stacking layers regularly resulted in prediction models that were either equally good or better than other prediction models based on other algorithms (see the bootstrapped differences in Figs. S19-S28). For instance, using Elastic Net across both layers for "Stacked: All" led to r at mean (M) = 0.60 (95% CI [0.56, 0.64]), M = 0.61 (95% CI [0.56, 0.66]), and M = 0.55 (95% CI [0.49, 0.60]) for HCP Young Adults, HCP Aging, and Dunedin Study, respectively.

Among the nonstacked models that predicted cognitive abilities at the time of scanning, we found varied predictive performance associated with different sets of MRI features. On the one hand, Task Contrasts from certain tasks led to top-performing models across the three datasets: the working-memory task in HCP Young Adults and the facename task in HCP Aging and Dunedin Study. With Elastic Net, these three task contrasts led to r at M=0.5 (95% CI [0.45, 0.59]) for HCP Young Adults, M= 0.46 (95% CI [0.39, 0.51]) for HCP Aging, and M=0.43 (95% CI [0.37, 0.49]) for Dunedin Study. On the other hand, Task Contrasts from some other tasks led to poor-performing models, such as the gambling task in HCP Young Adults (r could not be calculated due to the models resulting in the same predicted values on certain folds), the Conditioned Approach Response Inhibition Task (CARIT) task in HCP Aging, r at M = 0.07 (95% CI [-0.02, 0.15]), and the monetary incentive delay (MID) task in Dunedin Study, r at M = 0.16 (95% CI [0.08, 0.22]).

For Dunedin Study, the prediction models that predicted cognitive abilities at the time of scanning (i.e. when participants were 45 years old) performed similarly to those that predicted cognitive abilities, collected much earlier than the scanning time (i.e. when they were 7, 9, and 11 years old). For instance, the "Stacked: All" models using Elastic Net across both nonstacking and stacking layers predicted cognitive abilities at 45 years old and at 7, 9, and 11 years old with r at M = 0.55 (95% CI [0.49, 0.60]) and M =0.52 (95% CI [0.47, 0.57]), respectively. And Task Contrasts from the facename task led to top-performing models across two time points: with r at M = 0.43 (95% CI [0.36, 0.48]) at 7, 9, and 11 years old, compared with M = 0.43 (95% CI [0.37, 0.49]) at 45 years old.

In contrast, the performance of models predicting the residual scores for cognitive abilities from multimodal MRI was much poorer. Note that the negative residual scores reflect a stronger decline in cognitive abilities, as expected from childhood cognitive abilities, compared with participants' peers. The highest performing model



Fig. 2. Predictability of stacked and nonstacked models. a) Pearson's correlation (r) of stacked and nonstacked models for each dataset with Elastic Net across the two layers. Higher is better. Each dot represents predictive performance at each outer-fold test set. For other algorithms and other performance indices, the coefficient of determination (R²), and MAE, see Figs. S1–S9. For Dunedin Study, childhood scores reflect cognitive abilities, averaged across 7, 9, and 11 years old, and negative residual scores reflect a stronger decline in cognitive abilities, as expected from childhood cognitive abilities, compared with participants' peers. b) Dense scatter plot illustrating observed and predicted cognitive abilities (Z scores) using Stacked-All models with Elastic Net across two layers. Stacked All include all sets of MRI features. c) Observed cognitive abilities at ages 7, 9, and 11 years compared with age 45 years from the Dunedin Study. The ICC reflects the strength of the relationship in the observed cognitive ability scores between these time points. d) Predicted cognitive abilities at ages 7, 9, and 11 years of the velocities at ages 7, 9, and 11 years of the relationship in the predicted cognitive ability scores at each of the two time points were trained from the same set of neuroimaging features via the Stacked-All models, albeit with different targets (either the cognitive abilities at age 45 years). This is because MRI data were only collected at ge4 5 years, while cognitive abilities will be higher than the Pearson's correlation of the predicted cognitive ability scores. XGB, XGBoost.

predicting the residual scores across algorithms was the model with the encoding vs. distractor contrast from the facename task, followed by various stacked models. With Elastic Net, the best model that predicted the residual scores led to r at M = 0.21 (95% CI [0.14, 0.27]). And with Elastic Net across both layers for "Stacked: All" led to r at M = 0.17 (95% CI [0.10, 0.24]). While these r levels were statistically better than chance according to bootstrapping (see Fig. S10), it was much lower than those from prediction models that predicted cognitive abilities at the time of scanning or much younger age.

To understand the contribution of each MRI feature, we examined feature importance of each model based on Elastic Net coefficients. Given that Elastic Net features are linear and additive, the linear combination of Elastic Net coefficients reflects how the algorithm makes prediction. For stacked models, Fig. S29 shows the feature importance of stacked models for each dataset when predicting cognitive abilities at the time of scanning. The topperforming Task Contrasts contributed stronger in the stacked models across the three datasets. For nonstacked models, S30 and S31 show the feature importance of for each MRI modality, study, and target variable. Figure 3 and Tables S1–S10 show the feature importance of the top-performing, nonstacked models for each study and target variable.

Note that we provided tables of the numerical values of the predictability indices on our GitHub page: https://github.com/HAMlab-Otago-University/Predictability-Reliability-Generalizability/ tree/main/4_Supplementary.

Test-retest reliability

Figure 4 shows the test–retest reliability. Here, we tested the rank stability of predicted values from prediction models across two sessions, as indicated by interclass correlation (ICC). Given the availability of the test–retest participants, we examined the test–retest reliability of the stacked and nonstacked models from HCP Young Adults and Dunedin Study. We provided predicted values across two scanning sessions for each participant in Figs. S33 and S34 and ICC for each MRI feature before prediction modeling in Figs. S35–S38. Overall, for both datasets, prediction models with structural MRI, including total brain and subcortical volume, surface area, and cortical thickness, led to the highest level of test–retest reliability.

Similar to predictability, combining different sets of MRI features into stacked models mostly gave rise to high test-retest reliability. "Stacked: All," which included all sets of MRI features, resulted in an excellent ICC at 0.79 and 0.89 for HCP Young Adults and Dunedin Study, respectively. Moreover, we also found the boosting effect of stacking even when only fMRI during tasks was included in the models. For instance, "Stacked: Task Contrast and Task FC," which included the contrasts and FC from all of the fMRI tasks within each dataset, led to an excellent ICC at 0.8 and 0.87 for HCP Young Adults and Dunedin Study, respectively. Similarly, "Stacked: Task Contrast," which included the contrasts, but not FC, from all of the fMRI tasks within each dataset, still led to an excellent ICC at 0.77 and 0.77 for HCP Young Adults and Dunedin Study, respectively.

Among the nonstacked models that predicted cognitive abilities from fMRI (including Task Contrasts, Task FC, and Rest FC), some models showed a good-to-excellent level of ICC. These include a contrast from the language task (ICC = 0.77) and FC during rest (ICC = 0.77) in HCP Young Adults and FC during the MID task (ICC = 0.72) and rest (ICC = 0.63) in Dunedin Study. Yet, some models from fMRI provided a poor level of ICC, including a contrast

during the motor task (ICC = 0.38) in HCP Young Adults and FC and a contrast during the MID (ICC = 0.35) and Stroop (ICC = 0.28) tasks and FC (ICC = 0.24) during the emotion processing and facename tasks in Dunedin Study.

Generalizability

Figure 5 shows the generalizability among the three datasets. Here, we tested the performance of the prediction models trained from one dataset in predicting the cognitive abilities of participants from another separate dataset, as indicated by Pearson's correlation (r) between predicted and observed cognitive abilities. Given the different fMRI tasks used in different datasets, we only examined the generalizability of the prediction models using non-task sets of features (including rest FC, cortical thickness, cortical surface area, subcortical volume, total brain volume, and their combination, or "Stacked: Non Task"). Note that even the fMRI tasks with the same name, "the facename task" and "face or emotion processing" tasks were implemented differently across different datasets (see Materials and methods).

The "Stacked: Non Task" model showed generalizability at M = 0.25 (SD = 0.06). This level of cross-dataset generalizability was significantly better than chance (see the 95% CI in Fig. 5) and was similar to, albeit numerically smaller than, the within-dataset predictability of the models built from nested CV (M = 0.4, SD = 0.05). There were some differences in generalizability among pairs of studies. Generalizability was numerically higher between HCP Young Adults and HCP Aging (M = 0.33, SD = 0.04), as compared to between Dunedin Study and the other two datasets (M = 0.22, SD = 0.03).

Similarly, the nonstacked models with nontask sets of features showed generalizability at M = 0.18 (SD = 0.05). Apart from cortical thickness, the generalizability of every other nontask set of features was significantly better than chance (see the 95% CI in Fig. 5). Additionally, this level of cross-dataset generalizability from the nontask sets of features was similar to the within-dataset predictability of the models built from nested CV (M = 0.25, SD = 0.05).

We also examined the similarity in predicted values among the three datasets. Here, we tested the Pearson's correlation (r) in predicted values between the prediction models built from the same dataset and those built from another dataset (Fig. 5). The "Stacked: Non Task" model showed similarity in predicted values at M = 0.38 (SD = 0.12). This level of similarity was significantly better than chance (see the 95% CI in Fig. 5). As for the nonstacked models with nontask sets of features, we found the similarity in predicted values on average at M = 0.57 (SD = 0.11) and all of them were significantly better than chance (see the 95% CI in Fig. 5). Yet, some nontask sets of features appeared to be stronger in similarity than others. For instance, the similarity in predicted values for the total brain volume (M = 0.92, SD = 0.04) and subcortical volume (M = 0.84, SD = 0.09) was numerically higher than those for rest FC (M = 0.29, SD = 0.10), cortical area (M = 0.55, SD = 0.09), and cortical thickness (M = 0.26, SD = 0.22).

Discussion

Here, we examined stacking as a potential solution for improving BWAS for cognitive abilities in three key aspects: predictability, test-retest reliability, and generalizability (16, 20, 21). Most BWASs use one single modality of MRI to build prediction models, but here, we drew information across different MRI modalities via stacking (26). Stacked models demonstrated improvement in all



Fig. 3. Feature importance of the top-performing nonstacked models with Elastic Net, as indicated by Elastic Net coefficients. We grouped brain ROIs from the Glasser atlas (67) into 13 networks based on the Cole-Anticevic brain networks (66). In each figure, the networks are ranked by the mean Elastic Net coefficients, with the rankings shown to the right of each figure. The network partition illustration is sourced from the Actflow Toolbox https:// colelab.github.io/ActflowToolbox/. We provide actual values of the feature importance in Tables S1–S10.

three aspects and performed better than any individual modality in isolation. For predictability, stacked models led to high predictability, relative to what has been reported in the literature, across the three datasets when predicting cognitive abilities at the time of scanning. Notably, using the Dunedin Study, we were able to predict participants' cognitive abilities at ages 7, 9, and 11 years using their multimodal MRI at age 45 years, relatively well (r = 0.52). We found that this predictive performance was driven by



Test-Retest Reliability

Fig. 4. Test-retest reliability of the predicted values of the stacked and nonstacked models, indicated by ICC for HCP Young Adults and Dunedin Study. Left panel: Each dot represents ICC, while each bar represents a 95% CI. Right panel: Predicted values of some stacked models across two scanning sessions. Each line represents each participant. Lines would be completely parallel with each other in the case of perfect test-retest reliability. For other stacked and nonstacked models, see Figs. S33 and S34.

task contrasts, followed by task connectivity, across the three datasets. For test-retest reliability, stacked models reached an excellent level of reliability across HCP Young Adults and Dunedin Study, even when we only included fMRI during tasks in the models. For generalizability, combining nontask MRI features into a stacked model led to models that were applicable to other datasets, giving a level of performance that is better than chance. Altogether, the results optimistically support stacking as a viable approach to address the three challenges of BWAS for cognitive abilities.

Stacking improved predictability

Combining MRI across different sets of features via stacking consistently and substantially improved predictability within each dataset. Indeed, we found this improved predictability across three large-scale datasets that varied in age, culture, scanner manufacturer, scanning parameters, and cognitive ability assessments (27–29). This is consistent with our previous findings (24, 25). For cognitive abilities at the time of scanning, stacked models with all MRI sets of features led to *r* up to around 0.6, higher than those of nonstacked models in the current study, as well as those reported in a recent meta-analysis (r = 0.42 with 95% CI [0.35, 0.50]) (15). As reviewed in the meta-analysis (15), most studies applied similar machine learning algorithms to those used here but relied on a single MRI modality, most commonly FC during rest or task, followed by structural MRI. The use of task-fMRI contrasts is less common (39). Our nonstacked models with task FC, rest FC, and structural MRI (sMRI) showed similar performance to previous findings. Importantly, our stacked models that included Task Contrasts along with other modalities showed a much higher predictive performance (e.g. r = 0.604, $R^2 = 0.352$ based on "Stacked: All" with Elastic Net across both layers done on the HCP Young Adults), compared with the stacked models that did not include task fMRI in the current study as well as to the models in a previous study that also modeled the data from HCP Young Adults ($R^2 = 0.078$) (37). This confirms (i) that different MRI sets of features provide independent but complementary information about individual differences in cognitive abilities and (ii) that Task Contrasts, which are often ignored, could significantly help improve the predictive performance.

The superior predictability of Task Contrasts from certain tasks, especially when predicting cognitive abilities at the time of scanning, was also consistent with previous work (24, 25, 39,



Fig. 5. Generalizability and similarity in predicted values among the three datasets, as indicated by Pearson's correlation, *r*. Note that due to the different tasks used in different datasets, we only examined the generalizability of prediction models built from nontask sets of features (including rest FC, cortical thickness, cortical surface area, subcortical volume, total brain volume, and their combination, or "Stacked: Non Task"). For generalizability, the off-diagonal values reflect the level of generalizability from one dataset to another, while the diagonal values reflect the predictability of the models built from the same dataset via nested CV. For the similarity in predicted values, the off-diagonal values reflect the level of similarity in predicted values between two datasets. Higher values are better. The values in square blankets reflect a bootstrapped 95% CI. If 95% CI did not include 0, then generalizability/similarity in predictive values was better than chance. HCP-YA, HCP Young Adults; HCP-A, HCP Aging; DUD, Dunedin Study.

50). The working-memory task in HCP Young Adults and the facename task in HCP Aging and Dunedin Study created nonstacked models with the highest predictability for each dataset. The more popular MRI modalities, Task FC and Rest FC (15), did not perform as strongly as the Task Contrasts from working-memory and facename tasks. And structural MRI seemed to provide much poorer predictability across datasets, consistent with earlier work (49). It is important to note that not all Task Contrasts produced prediction models with high predictability. In fact, the worst prediction models across the three studies were also Task Contrasts (e.g. the gambling, CARIT and MID tasks in HCP Young Adults, HCP Aging, and Dunedin Study, respectively). This suggests the selectivity of the fMRI BOLD activity relevant to events for different tasks-some tasks were related to individual differences in cognitive abilities and some tasks were not. The best tasks here were related to either working memory or episodic memory, which might reflect what was being measured with the cognitive ability assessments (17, 63), i.e. through NIH Toolbox (64) or Wechsler Adult Intelligence Scale (WAIS) (65). Accordingly, to further improve the predictability of BWAS for cognitive abilities via task contrasts, future research will need to determine which tasks are more relevant to cognitive abilities. If the tasks used have not yet been established to predict cognitive abilities, researchers may

still consider stacking to determine whether combining Task Contrasts from such tasks with other MRI modalities improves predictability. By using multivariate predictive modeling algorithms, such as Elastic Net, to stack different Task Contrasts, tasks that do not enhance predictability will have less weight in the final stacked model due to regularization. This makes it safe to include any available tasks. Researchers may then decide to drop the less contributing tasks to create a more parsimonious model.

We also examined the predictability of stacked models in light of the Dunedin Study's longitudinal measurements for cognitive abilities (28). Stacked models with all MRI sets of features were able to predict cognitive abilities, collected 36 years before the scanning time, at a similarly high level of performance to those at the time of scanning (r = 0.55 vs. r = 0.52, respectively). Yet, when predicting the residual scores, the stacked models with all MRI sets of features gave much lower predictability, albeit still significant, at r = 0.17. These residual scores reflect changes in cognitive abilities from childhood to middle age, compared with participants' peers. This pattern of results may suggest that brain information revealed by multimodal MRI, obtained in the middle age (i.e. 45 years old), is more related to the stable trait of cognitive abilities, but less to the changes over 35 years. Perhaps, this is because individual differences in cognitive abilities were stable over the lifespan (1), making it easier for multimodal MRI to capture their intervariability over intraindividual variability. Using the Dunedin Study, we indeed found a high rank stability of this trait: that childhood cognitive abilities were related to middleaged cognitive abilities at ICC = 0.78 and that the multimodal MRI predicted values of cognitive abilities based on either time points led to very similar scores at r = 0.94 (see Fig. 2c and d). Accordingly, if the aim of BWAS is to capture the stable trait of cognitive abilities, the current approach of stacking multimodal MRI data from one time point seems appropriate. It is important to note that our work only demonstrates the retrospective prediction of cognitive abilities (i.e. using MRI at age 45 years to predict cognitive abilities 36 years prior). Future work is needed to examine whether our method can be extended to forecast cognitive abilities. Fortunately, the Dunedin Study is still ongoing, and we hope to test whether our method can use MRI at age 45 years to predict future cognitive decline as well as cognitive-related neurological disorders, such as mild cognitive impairments, dementia, and Alzheimer's, as the participants age.

To explain how each model made predictions, we treated Elastic Net coefficients as indicators of feature importance. Examining the feature importance in the stacked models revealed that the topperforming modality, specifically the best Task Contrasts, was the strongest contributor. This pattern was consistent across datasets. This suggests that the top-performing Task Contrasts, such as the working-memory task (reflected by high over lower workingmemory load conditions) in HCP Young Adults and the facename task (reflected by encoding over distractor/control conditions) in HCP Aging and the Dunedin Study, provided strong and unique contributions to the overall prediction of the stacked models. Examining the feature importance of these top-performing Task Contrasts illustrates the contribution from each brain area.

We grouped brain areas into 13 different networks based on the Cole-Anticevic definition (66). Contributions from different brain networks varied depending on the specific Task Contrasts, datasets, and cognitive target variables. Nonetheless, some patterns emerged. For instance, our prediction models indicated that participants with more positive Task Contrasts from brain areas within the default mode network tended to have worse cognitive abilities. Conversely, participants with more positive Task Contrasts from brain regions within the dorsal attention network tended to have better cognitive abilities at the time of scanning, cognitive abilities 35 years prior, and residual scores. Accordingly, we demonstrated the contribution of certain networks within the context of specific tasks in predicting cognitive abilities.

Stacking improved test-retest reliability

Creating prediction models from separate MRI sets of features and combining them via stacking also improved reliability for Dunedin Study (28), especially for Task Contrasts, similar to our previous findings (25) with HCP Young Adults (29). This approach, in effect, addresses the poor reliability of Task Contrasts, found earlier in the same two datasets (20). That is, previous work (20) focused on the reliability of Task Contrasts at specific brain areas from certain tasks and found low test–retest reliability (also demonstrated here in Figs. S35 and S36). Here, instead of focusing on specific areas, we used multivariate methods and stacking to draw information across the whole brain and tasks and found a boost in reliability. Indeed, stacked models that combined only Task Contrasts and that combined Task Contrasts and FC together both gave the ICC at excellent levels (i.e. ICC \geq 0.75) across the two datasets.

While the prediction models from structural MRI sets of features, e.g. surface area, total brain volume, subcortical volume, led to the highest level of test-retest reliability, these models provided poorer predictability for cognitive abilities. This high level of test-retest reliability from structural MRI is not surprising since we should not expect drastic changes in brain anatomy in a short period of time, assuming no major brain incidents (e.g. concussion or stroke) (20). In contrast, the stacked models from Task Contrasts and FC also provided an excellent level of test-retest reliability (albeit not as high as structural MRI models), but they gave much higher predictability. Accordingly, future BWAS for cognitive abilities that would like to optimize both reliability and predictability might prefer stacking Task Contrasts and FC, or better yet stacking all the MRI data available, rather than relying on structural MRI.

Stacking of non-task MRI sets of features led to better-than-chance generalizability

Unlike predictability and reliability, we could only focus on the nontask MRI sets of features (including rest FC, cortical thickness, cortical surface area, subcortical volume, and total brain volume) for cross-cohort generalizability, given the differences in fMRI tasks used in each dataset. We found that the "Stacked: Non Task" models, built from one dataset, predicted the cognitive abilities of the participants in the other two datasets better than chance. Still, if we treated the within-dataset predictability as the ceiling of cross-dataset generalizability, the cross-dataset generalizability of the "Stacked: Non Task" models (r = 0.25) was numerically lower than the ceiling (r = 0.4).

One caveat is that the generalizability of the "Stacked: Non Task" models between HCP Young Adults and HCP Aging (r= 0.33) is numerically higher than those between Dunedin Study and the two HCP datasets (r = 0.22). Consistent with this is the numerically higher similarity in predicted values between HCP Young Adults and HCP Aging (r = 0.51) compared with between Dunedin Study and the two HCP datasets (r = 0.32). This may reflect a higher homogeneity between the two HCP datasets. While the two HCP datasets differed in the age of participants and certain scanning parameters (e.g. repetition time [TR]), HCP Aging (27) was modeled after the earlier success of HCP Young Adults (29). The two HCP datasets, for instance, used the NIH Toolbox (64) to access cognitive abilities, while Dunedin Study (28) used WAIS (65). Nonetheless, testing generalizability on Dunedin Study that was conducted independently from the HCPs may provide a more realistic picture of how deployable the "Stacked: Non Task" models to indicate cognitive abilities in the real world.

As for the nonstacked models, cross-cohort generalizability was mostly significant, except for cortical thickness. This is in line with previous studies focusing on cross-dataset generalizability of Rest FC (55–57). The generalizability of structural MRI sets of features was more varied. Some structural MRI sets gave generalizability close to predictability and provided high similarity in predicted values: total brain volume (generalizability = 0.24, predictability = 0.23, and similarity = 0.92) and subcortical brain volume (generalizability = 0.21, predictability = 0.22, and similarity = 0.84). But other structural MRI sets did not: cortical areas (generalizability = 0.16, predictability = 0.27, and similarity = 0.55) and cortical thickness (generalizability = 0.10, predictability = 0.23, and similarity = 0.26). It is hard to pinpoint whether this is due to the differences in scanning parameters between datasets or, instead, due to the nature of the sets of features. Future research with a larger number of datasets is needed to pinpoint the characteristics of the datasets and/or features that could lead to better generalizability.

Limitations and future directions

The current study has several limitations. First, it would have been desirable to examine the generalizability of stacked models involving Task Contrasts and Task FC in all cohorts. Stacked models involving Task Contrasts and FC scored high in both predictability (especially when compared with "Stacked: Non Task" models) and reliability. The inability to test their generalizability means that we cannot know for sure how deployable these highly predictable models are. Thus, for the time being, we advise researchers who would like to apply the stacked models with task fMRI on new data to follow the procedures of the original datasets as much as possible. This could be task design and scanning parameters among others.

Second, we mainly relied on the fMRI tasks and preprocessing pipelines chosen by the original investigators of each dataset. However, the fMRI tasks they chose might not be optimized for predictability, reliability, and generalizability for cognitive abilities. As suggested elsewhere (52), perhaps fMRI tasks need to be designed from the ground up, using tools such as item response theory, to ensure that they capture individual differences well. Fortunately, some of the fMRI tasks (e.g. the working-memory and language tasks) provided relatively high predictability and reliability for cognitive abilities. Based on our results, in a situation where optimization of the tasks is unknown, stacking Task Contrasts and Task FC across all of the available fMRI tasks should provide the best performance possible, given the choice of the tasks used. Similarly, each dataset's preprocessing approach might not be optimized for BWAS with cognitive abilities as a target. For instance, for Rest FC in HCP Young Adults, we treated a choice of the two denoising strategies as another hyperparameter to select from the training sets: the investigators' recommended method ICA-FIX (67) and an alternative method aCompCor (68). We found that aCompCor (68) performed better in the training sets across different prediction algorithms (see Fig. S32), despite not being used in the original preprocessing pipeline (29, 69, 70). While using the recommended preprocessing pipeline for each dataset allowed for easier reproducibility, we still need to test whether predictability, reliability, and generalizability could be further improved with more refined pipelines, optimized for predicting cognitive abilities.

Third, while predictability, reliability, and generalizability are important for multimodal MRI to be applied as a neural indicator for cognitive abilities (21), other aspects still need to be accomplished for cognitive neuroscientists to truly understand the relationship between cognitive abilities and multimodal MRI measures. For instance, to reveal how the prediction models draw information from each MRI set of features, we need prediction models with good explainability (17, 71). Yet, the current prediction models are optimized for predictability, but not explainability. We previously proposed several methods to improve the explainability of prediction models (17). For example, to provide statistical inference for feature importance, future researchers could create a null distribution of feature importance via permutation, allowing them to determine whether the contribution of each feature is significantly better than chance, a technique called eNetXplorer (17, 72). Similarly, to demonstrate the pattern (i.e. linearity vs. nonlinearity) and directionality (i.e. positive vs. negative) of the relationship between each MRI feature and the prediction, future researchers could apply a visualization technique called Accumulated Local Effects (ALE) (17, 73). Lastly, for interactive algorithms (e.g. XGBoost and Random Forest), future researchers could use Friedman's H-statistic (17, 74) to quantify the interaction strength between each MRI feature and all other MRI features in making the prediction. However, optimizing explainability is beyond the scope of this study.

Conclusion

Cognitive neuroscientists have long dreamt of the ability to associate individual differences in cognitive abilities with brain variations (10). Yet, BWASs need to be improved in their predictability, test-retest reliability, and generalizability before they can produce a robust neural indicator for cognitive abilities (16, 20, 21, 75, 76). Based on our benchmark, combining different modalities of MRI into one prediction model via stacking seems to be a viable approach to realize this dream of cognitive neuroscientists.

Materials and methods

Datasets

In this study, we analyzed three datasets with 2,131 participants (1,139 females) in total (see Fig. 1 for their age distribution; see Supplementary Methods for participants' details). These three datasets have been approved by the institutional review boards at the institutions where the data were collected, and informed consent was obtained (see references (27, 29, 77)). The current authors applied for and received authorization from the original investigators to conduct secondary analyses on these datasets. As these datasets are deidentified and designated for public use, the ethics committee at University of Otago determined that this research does not constitute human subjects research, and therefore, ethical approval was not required.

Test–retest subsets

HCP Young Adults and Dunedin Study had a subset of participants who completed the entire MRI procedure twice. In HCP Young Adults, 45 participants were scanned M = 139 (SD = 67.3) days apart, and the exclusion criteria left 34 participants. In Dunedin Study, 20 participants were scanned M = 79 (SD = 10.4) days apart.

Features: multimodal MRI

We used the following MRI modalities: task-fMRI contrasts, task-fMRI FC, resting-state fMRI connectivity, and structural MRI.

Task-fMRI contrasts (task contrasts)

Task Contrasts reflect fMRI BOLD activity relevant to events in each task. We used Task Contrasts, preprocessed by each of the three studies. See Supplementary methods for the details about task-contrast features. Briefly, for each study, we extracted one set of 379 (i.e. region of interests [ROIs]) per contrast, leaving seven, five, and four sets of 379 task-contrast features for HCP Young Adults, HCP Aging, and Dunedin Study, respectively.

Task-fMRI FC

Task FC reflects FC during each task. Studies have considered task FC as an important source of individual differences (53, 78, 79). As opposed to creating contrasts from fMRI time series during each task as in Task Contrasts, here we computed FC, controlling for HRF-convolved events from each task. See Supplementary methods for the details about Task FC features. Briefly, for each study, we extracted one set of 75 task FC features (principal components) per contrast, leaving seven, five, and four sets of 75 task FC features for HCP Young Adults, HCP Aging, and Dunedin Study, respectively.

Resting-state fMRI FC

Rest FC reflects FC during rest. Both HCP Young Adults and HCP Aging included four runs of rest FC, each at 14:33 min and 6:42 min long, respectively. Dunedin Study included only one run of rest FC with 8:16 min long. See Supplementary methods for the details about Rest FC features. Briefly, we obtained one set of 75 rest FC features (principal components) for each of the three datasets.

Structural MRI

Structural MRI reflects individual differences in brain anatomy. The three studies applied Freesurfer (80) to quantify these individual differences. Here, we focused on four sets of features: cortical thickness, cortical surface area, subcortical volume, and total brain volume. Specifically, for cortical thickness and cortical surface area, we created 148 vertex-based ROIs using the Destrieux atlas (81), while for subcortical volume, we created 19 voxel-based ROIs using the ASEG atlas (80). For total brain volume, we used summary indices provided by Freesurfer(80). See Supplementary Methods for details.

Target: cognitive abilities

Cognitive abilities were measured outside of the MRI. HCP Young Adults and HCP Aging measured cognitive abilities using the NIH Toolbox (64). Here, we used a summary score (CogTotalComp_ Unadj) that covered behavioral performance from several tasks, including picture sequence memory, Flanker, list sorting, dimensional change card sort, pattern comparison, reading tests, and picture vocabulary.

Dunedin Study measured cognitive abilities in several visits. We computed three scores and used them as separate targets. The first score is cognitive abilities, collected as part of the MRI visit at 45 years old via the WAIS-IV scale (65). The second score is cognitive abilities, averaged across 7, 9, and 11 years old, collected using the Wechsler Intelligence Scale for Children—Revised (82). The third score is the residual scores for the cognitive abilities (62), calculated as follows. We, first, used linear regression to predict cognitive abilities at 45 years old from cognitive abilities at 7, 9, and 11 years old. We, then, subtracted the predicted values of this linear regression from the actual cognitive abilities at 45 years old, creating the residual cognitive abilities. Negative scores of these residual cognitive abilities reflect a stronger decline in cognitive abilities, as expected from childhood cognitive abilities, compared with participants' peers. Note that, due to the differences in the cognitive measures used for age 45 vs. ages 7, 9, and 11 years, we could not simply subtract the scores between the two time points. While using the residual scores did not provide a change in cognitive abilities in absolute terms, they still indicate the relative changes in an individual's cognitive abilities compared with their peers. See Prediction models below for our approach to prevent data leakage when calculating this residual score.

Prediction models

Similar to our previous work(25), we employed nested CV to predict cognitive abilities from multimodal MRI data (see Fig. 1). Initially, we divided the data from each study into outer folds. The number of outer folds was determined to ensure at least 100 participants per fold. Consequently, we had eight outer folds for HCP Young Adults, five for HCP Aging, and seven for the Dunedin Study. For HCP Young Adults, which included participants from the same families, we created the eight outer folds based on ~50 family groups, ensuring that members of the same family were in the same outer fold.

We then iterated through the outer folds, treating one fold as the test set and the remaining folds as the training set. This approach resulted in around 100 participants in each outer-fold test set for all three studies, with ~700, 400, and 600 participants in the outer-fold training sets for HCP Young Adults, HCP Aging, and the Dunedin Study, respectively. Next, we split each outerfold training set into five inner folds. We iterated through these inner folds to tune the hyperparameters of the prediction models, selecting the final models based on the coefficient of determination (R²), a default option in sklearn. To prevent data leakage between the outer-fold training and test sets when calculating residual scores for cognitive abilities in the Dunedin Study, we created linear regression models to predict cognitive abilities at age 45 years from abilities at ages 7, 9, and 11 years using the outerfold training set and applied these models to the corresponding outer-fold test set.

Apart from training each of the sets of multimodal MRI features to predict cognitive abilities in separate prediction models, known as "nonstacked" models, we also combined different sets together via stacking, creating "stacked" models (see Fig. 1). To train the stacked models, once we finished training all of the nonstacked models from every set of features, we computed predicted values from these nonstacked models. Specifically, we used only the data from each outer-fold training set to train the stacked models and treated the predicted values from the nonstacked models as features to predict cognitive abilities. For example, to create a stacked model to combine Task Contrasts for HCP Young Adults, we first created nonstacked models for each of the seven sets of 379 task-contrast features available in this dataset. Then, we computed the predicted values of each of these seven nonstacked models, using them as seven features in a stacked model that was trained to predict cognitive abilities in the outerfold training sets. We tuned these stacked models using the same inner-fold CV as the nonstacked models. Accordingly, the training of stacked and nonstacked models did not involve outer-fold test sets, preventing data leakage. Note that by creating just one predicted value per nonstacked model, our stacking approach is sometimes called "late fusion," which differs from simply concatenating features from different sets, known as "early fusion" or "flat model" (25, 83). One benefit of stacking predicted values over concatenating features is the ability to use different machine learning algorithms within and across sets of features.

We created eight stacked models: "Task Contrast" including Task Contrasts from all of the tasks, "Task FC" including Task FC from all of the tasks, "Non Task" including Rest FC and structural MRI, "Task Contrast & FC" including task contrasts and task FC from all of the tasks, "All" including all sets of features, "All excluding Task Contrast" including every set of features except Task Contrasts, "All excluding Task FC" including every set of features except Task FC, and "Resting and Task FC" including FC during rest and tasks. Note that we used two strategies, ICA-FIX(67) and aCompCor(68), to denoise Rest FC for HCP Young Adults. We ultimately picked aCompCor (68) to be included in the stacked models since it led to a better predictive performance in the outer-fold training sets (see Fig. S32).

We applied corrections to reduce the influences of potential confounds. First, for all three studies, we controlled for biological sex by residualizing biological sex from all MRI features and cognitive abilities. For HCP Young Adults and HCP Aging, we also controlled for age, in addition to biological sex, from all MRI features and cognitive abilities. We did not control for age in Dunedin Study because all of the participants were scanned and measured their cognitive abilities at roughly the same age. Additionally, we residualized motion (average of relative displacement, Movement_RelativeRMS_mean) from task contrasts for HCP Young Adults and Dunedin Study. We did not residualize motion from task contrasts for HCP Aging as well as task FC and rest FC for all studies since either ICA-FIX (67) or aCompCor (68) was already applied to each participant. We also standardized all MRI features. To avoid data leakage, we first applied all residualization and standardization on each outer-fold training set. We then applied the parameters of these residualization and standardization to the corresponding outer-fold test set.

As in our previous article (25), we implemented four multivariate, predictive modeling algorithms via Scikit-learn (84): Elastic Net (58), SVR (59, 85), Random Forest (60), and XGBoost (61). For stacked models, we needed to apply the algorithm to two layers: (i) nonstacked layer (Step 1, Fig. 1), or on each set of features and (ii) stacked layer (Step 2, Fig. 1), or on the predicted values from each set. Accordingly, we implemented 16 (i.e. four-by-four across two layers) combinations of algorithms for the stacked models. See the details about predictive modeling algorithms in Supplementary Methods.

Predictability

To evaluate the predictability of prediction models, we computed the predicted values of the models at each outer-fold test set and compared them with the observed cognitive abilities. We calculated three performance indices for predictability: Pearson's correlation (r), the coefficient of determination (\mathbb{R}^2), and mean absolute error (MAE). Note for \mathbb{R}^2 , we applied the sum of squares definition (i.e. $\mathbb{R}^2 = 1 - [sum of squares residuals/total sum of$ squares]) and not the square of <math>r, following a previous recommendation (31).

To quantify the uncertainty around these performance indices, we calculated bootstrapped 95% CI (86). Here, we combined predicted and observed cognitive abilities across outer-fold test sets, sampled these values with replacement 5,000 times, and computed the three performance indices each time, giving us a bootstrapped distribution for each index. If the 95% CI of the *r* or R^2 bootstrapped distribution was higher than 0, then the predictability from a particular prediction model was better than chance.

To compare predictability among prediction models, we also used the bootstrapping approach (86). Similar to the above, we sampled, with replacement for 5,000 times, the observed cognitive abilities along with their predicted values from different prediction models across outer-fold test sets. In each sample, we computed each performance index of each prediction model and subtracted this index from that of the prediction model with the highest predictability of each dataset. If the 95% CI of this distribution of the subtractions was higher than 0, then we concluded that the prediction model we tested had significantly poorer performance than the prediction model with the highest predictability. We applied this approach separately for nonstacked and stacked models, allowing us to evaluate the best nonstacked and stacked models for each dataset. To understand how the prediction models drew information across multimodal MRI features, we plotted Elastic Net coefficients. We chose Elastic Net coefficients because (i) Elastic Net led to high predictability, as high as or higher than other algorithms (see Results) and (ii) the Elastic Net coefficients are readily interpretable. Elastic Net creates a predicted value from a weighted sum of features, and therefore, a stronger magnitude of an Elastic Net coefficient means a higher contribution to the prediction.

Our use of nested CV led to separate Elastic Net models, one for each outer fold, making it hard to visualize Elastic Net coefficients across all participants in each dataset. To address this, we retrained Elastic Net using the whole data (i.e. without splitting the data into outer folds) in each dataset and applied five CVs to tune the model. We then plotted the Elastic Net coefficients on brain images using brainspace (87) and nilearn (88). Note that we modeled Task FC and Rest FC after reducing their dimension via PCA. To extract the feature importance at each ROI-pair index, we multiplied the absolute PCA scores with Elastic Net coefficients and then summed the multiplied values across the 75 components, resulting in 71,631 ROI-pair indices.

Test-retest reliability

Given the high predictability of Elastic Net (see Figs. S19–S28), we evaluated the test–retest reliability of the prediction models based on Elastic Net. To evaluate test–retest reliability, we used HCP Young Adults and Dunedin Study test–retest subjects (i.e. participants who were scanned twice) as the test set and the rest of the participants in each dataset as a training set. Within the training set, we used the same five CVs to tune the Elastic Net models, as described above. We then examined the test–retest reliability of the predicted values between the first and second MRI sessions, as quantified by ICC 3.1 (89) via pingouin (https://pingouin-stats. org/):

$$\frac{\mathrm{MS}_p - \mathrm{MS}_e}{\mathrm{MS}_p + (k-1)\mathrm{MS}_e},$$

where MS_p is mean square for participants, MS_e is the mean square for error, and k is the number of time points. We used the following criteria to interpret ICC (90): ICC < 0.4 as poor, ICC \geq 0.4 and < 0.6 as fair, ICC \geq 0.6 and < 0.75 as good, and ICC \geq 0.75 as excellent reliability.

Generalizability

Similar to test-retest reliability, we evaluated the generalizability of the prediction models based on Elastic Net given the high predictability of Elastic Net based on bootstrapped comparisons (see Figs. S19–S28). For features, because the three datasets used mostly different fMRI tasks, we focused on the generalizability of nontask sets of features (including rest FC, cortical thickness, cortical surface area, subcortical volume, and total brain volume) and their stacked model, "Stacked: Non Task." For the target, we standardized cognitive abilities using a Z-score within each dataset, so that the target for each dataset was at the same standardized scale before model fitting. This is because Dunedin Study used WAIS-IV for measuring cognitive abilities, while Human Connectome Projects—Young Adults (HCP Young Adults) and Human Connectome Projects-Aging (HCP Aging) used NIH toolbox (64). Note that, for Dunedin Study, we only focused on cognitive abilities collected during the MRI visit at age 45 years (as opposed to during earlier visits) as the target, given that the other two studies only provided cognitive abilities during the MRI visit.

To evaluate generalizability across datasets, we treated one of the three datasets as a training set and the other two as two separate test sets. We computed the predicted values of the models at each test dataset and compared them with the observed cognitive abilities using Pearson's correlation (r). To examine whether the generalizability was statistically significant, we bootstrapped r 5,000 times. If the 95% bootstrapped CI was higher than 0, the r was statistically significantly better than chance. We also compared generalizability across datasets to predictability within each dataset using nested CVs. For predictability within each dataset, we combined predicted values across outer test sets and compared them with the observed cognitive abilities. We considered the predictability within each dataset as the ceiling of how high generalizability across datasets could be.

To further understand the extent to which the prediction models built from one dataset are different from those built from another, we also examined the similarity between predictive values. Here, using Pearson's correlation (r), we compared the correlation in predictive values from the prediction models built from the same dataset and those built from another dataset. Similar to generalizability, to test whether the similarity between predictive values was statistically significant, we bootstrapped r 5,000 times. If the 95% bootstrapped CI was higher than 0, the r was statistically significantly better than chance.

Acknowledgments

The authors thank the Dunedin Study members, their families, and friends for their long-term involvement. The authors also thank Dunedin Unit Director, Professor Reremoana (Moana) Theodore, Unit research staff, Professor Terrie E. Moffitt, Professor Avshalom Caspi, previous Study Director, Emeritus Distinguished Professor, the late Richie Poulton, for his leadership during the Study's research transition from young adulthood to aging (2000–2023) and Study founder, Dr Phil A. Silva. Professor Terrie E. Moffitt was especially helpful in providing comments on earlier drafts. The authors wish to acknowledge the use of New Zealand eScience Infrastructure (NeSI) high-performance computing facilities, consulting support, and/or training services as part of this research.

Supplementary Material

Supplementary material is available at PNAS Nexus online.

Funding

The Dunedin Multidisciplinary Health and Development Research Unit was supported by the Health Research Council of New Zealand and New Zealand Ministry of Business, Innovation and Employment. Funding support was also received from the New Zealand Health Research Council Programme grant (16-604), US-National Institute on Aging grants AG049789-06 and R01AG032282-11, and UK Medical Research Council grant MR/ X021149/1. Additional data were provided by the HCP Young Adults and the HCP Aging. The HCP Young Adults, WU-Minn Consortium (Principal Investigators: David Van Essen and Kamil Ugurbil; 1U54MH091657) was funded by the 16 NIH Institutes and Centers that support the NIH Blueprint for Neuroscience Research and by the McDonnell Center for Systems Neuroscience at Washington University. The HCP Aging was supported by the National Institute on Aging of the National Institutes of Health under award number U01AG052564 and by funds provided by the

McDonnell Center for Systems Neuroscience at Washington University in St. Louis. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health. New Zealand's national facilities were provided by New Zealand eScience Infrastructure (NeSI) and funded jointly by NeSI's collaborator institutions and through the Ministry of Business, Innovation and Employment's Research Infrastructure Programme. URL https://www.nesi.org.nz. A.T., T.M., and N.P. were supported by Health Research Council of New Zealand (grant numbers 21/618 and 24/838), the University of Otago, and the Neurological Foundation of New Zealand (grant number 2350 PRG).

Preprints

This preprint on biorXiv is available under a CC-BY-NC 4.0 International license (https://doi.org/10.1101/2024.05.03.589404).

Data Availability

All codes are available at https://github.com/HAM-lab-Otago-University/Predictability-Reliability-Generalizability. Instructions for data access can be found here: HCP Young Adults https://www. humanconnectome.org/study/hcp-young-adult; HCP Aging https:// www.humanconnectome.org/study/hcp-lifespan-aging; and Dunedin Study https://dunedinstudy.otago.ac.nz/. While we did not preregister our data analysis plan for HCP Young Adults and HCP Aging, we preregistered our plan to test predictability and test-retest reliability for the Dunedin Study prior to having access to the dataset at https://dunedinstudy.otago.ac.nz/files/ 1639954373_Pat%20Multimodal%20brain%20concept%20paper_NP %20TEMsigned.docx.pdf.

References

- 1 Deary IJ, Pattie A, Starr JM. 2013. The stability of intelligence from age 11 to age 90 years: the Lothian Birth Cohort of 1921. Psychol Sci. 24:2361–2368.
- 2 Tucker-Drob EM, Briley DA. 2014. Continuity of genetic and environmental influences on cognition across the life span: a metaanalysis of longitudinal twin and adoption studies. *Psychol Bull.* 140:949–979.
- 3 Deary IJ, Strand S, Smith P, Fernandes C. 2007. Intelligence and educational achievement. *Intelligence*. 35:13–21.
- 4 Schmidt FL, Hunter J. 2004. General mental ability in the world of work: occupational attainment and job performance. J Pers Soc Psychol. 86:162–173.
- 5 Llewellyn DJ, Lang IA, Langa KM, Huppert FA. 2008. Cognitive function and psychological well-being: findings from a population-based cohort. Age Ageing. 37:685–689.
- 6 Strenze T. 2007. Intelligence and socioeconomic success: a metaanalytic review of longitudinal research. Intelligence. 35:401–426.
- 7 Calvin CM, et al. 2017. Childhood intelligence in relation to major causes of death in 68 year follow-up: prospective population study. BMJ. 357:j2708. https://doi.org/10.1136/bmj.j2708
- 8 East-Richard C, Mercier AR, Nadeau D, Cellard C. 2020. Transdiagnostic neurocognitive deficits in psychiatry: a review of meta-analyses. *Can Psychol.* 61:190–214.
- 9 Abramovitch A, Short T, Schweiger A. 2021. The C factor: cognitive dysfunction as a transdiagnostic dimension in psychopathology. Clin Psychol Rev. 86:102007.
- 10 Deary IJ, Penke L, Johnson W. 2010. The neuroscience of human intelligence differences. Nat Rev Neurosci. 11:201–211.

- 11 Sui J, Jiang R, Bustillo J, Calhoun V. 2020. Neuroimaging-based individualized prediction of cognition and behavior for mental disorders and health: methods and promises. *Biol Psychiatry*. 88: 818–828.
- 12 Morris SE, Cuthbert BN. 2012. Research domain criteria: cognitive systems, neural circuits, and dimensions of behavior. *Dialogues Clin Neurosci.* 14:29–37.
- 13 Horien C, et al. 2021. A hitchhiker's guide to working with large, open-source neuroimaging datasets. Nat Hum Behav. 5:185–193.
- 14 Woo C-W, Chang LJ, Lindquist MA, Wager TD. 2017. Building better biomarkers: brain models in translational neuroimaging. Nat Neurosci. 20:365–377.
- 15 Vieira BH, et al. 2022. On the prediction of human intelligence from neuroimaging: a systematic review of methods and reporting. Intelligence. 93:101654.
- 16 Marek S, et al. 2022. Reproducible brain-wide association studies require thousands of individuals. *Nature*. 603:654–660.
- 17 Pat N, Wang Y, Bartonicek A, Candia J, Stringaris A. 2023. Explainable machine learning approach to predict and explain the relationship between task-based fMRI and individual differences in cognition. *Cerebral Cortex*. 33:2682–2703.
- 18 Spisak T, Bingel U, Wager TD. 2023. Multivariate BWAS can be replicable with moderate sample sizes. *Nature*. 615:E4–E7.
- 19 Choi SW, Mak TS-H, O'Reilly PF. 2020. Tutorial: a guide to performing polygenic risk score analyses. Nat Protoc. 15:2759–2772.
- 20 Elliott ML, et al. 2020. What is the test-retest reliability of common task-functional MRI measures? New empirical evidence and a meta-analysis. Psychol Sci. 31:792–806.
- 21 Wu J, Li J, Eickhoff SB, Scheinost D, Genon S. 2023. The challenges and prospects of brain-based prediction of behaviour. *Nat Hum Behav.* 7:1255–1264.
- 22 Nature. 2022. Cognitive neuroscience at the crossroads. Nature. 608:647.
- 23 APS, Scanning the brain to predict behavior, a daunting 'Task' for MRI. Association for Psychological Science—APS, 2020. [accessed 2020 Sep 27]. https://www.psychologicalscience.org/news/releases/ scanningthe-brain-fmri.html.
- 24 Pat N, *et al.* 2022. Longitudinally stable, brain-based predictive models mediate the relationships between childhood cognition and socio-demographic, psychological and genetic factors. *Hum Brain Mapp.* 43:5520–5542.
- 25 Tetereva A, Li J, Deng JD, Stringaris A, Pat N. 2022. Capturing brain-cognition relationship: integrating task-based fMRI across tasks markedly boosts prediction and test-retest reliability. *NeuroImage*. 263:119588.
- 26 Engemann DA, et al. 2020. Combining magnetoencephalography with magnetic resonance imaging enhances learning of surrogate-biomarkers. eLife. 9:e54055.
- 27 Bookheimer SY, et al. 2019. The lifespan human connectome project in aging: an overview. *NeuroImage*. 185:335–348.
- 28 Poulton R, Moffitt TE, Silva PA. 2015. The Dunedin multidisciplinary health and development study: overview of the first 40 years, with an eye to the future. Soc Psychiatry Psychiatr Epidemiol. 50: 679–693.
- 29 Van Essen DC, et al. 2013. The WU-Minn Human Connectome Project: an overview. NeuroImage. 80:62–79.
- 30 Yarkoni T, Westfall J. 2017. Choosing prediction over explanation in psychology: lessons from machine learning. Perspect Psychol Sci. 12:1100–1122.
- 31 Poldrack RA, Huckins G, Varoquaux G. 2020. Establishment of best practices for evidence for prediction: a review. JAMA Psychiatry. 77:534–540.

- 32 Varoquaux G, et al. 2017. Assessing and tuning brain decoders: cross-validation, caveats, and guidelines. NeuroImage. 145: 166–179.
- 33 Genç E, et al. 2018. Diffusion markers of dendritic density and arborization in gray matter predict differences in intelligence. Nat Commun. 9:1905.
- 34 Narr KL, et al. 2007. Relationships between IQ and regional cortical gray matter thickness in healthy adults. Cerebral Cortex. 17:2163–2171.
- 35 Ohtani T, et al. 2014. Medial frontal white and gray matter contributions to general intelligence. PLoS One. 9:e112691.
- 36 Krämer C, et al. 2024. Prediction of cognitive performance differences in older age from multimodal neuroimaging data. *GeroScience*. 46:283–308.
- 37 Rasero J, Sentis AI, Yeh F-C, Verstynen T. 2021. Integrating across neuroimaging modalities boosts prediction accuracy of cognitive ability. PLoS Comput Biol. 17:e1008347.
- 38 Dubois J, Galdi P, Paul LK, Adolphs R. 2018. A distributed brain network predicts general intelligence from resting-state human neuroimaging data. Phil Trans R Soc B: Biol Sci. 373:20170284.
- 39 Sripada C, Angstadt M, Rutherford S, Taxali A, Shedden K. 2020. Toward a "treadmill test" for cognition: improved prediction of general cognitive ability from the task activated brain. *Hum Brain Mapp.* 41:3186–3197.
- 40 Sripada C, et al. 2021. Brain-wide functional connectivity patterns support general cognitive ability and mediate effects of socioeconomic status in youth. *Transl Psychiatry*. 11:571.
- 41 Gao S, Greene AS, Constable RT, Scheinost D. 2019. Combining multiple connectomes improves predictive modeling of phenotypic measures. *NeuroImage*. 201:116038.
- 42 Jiang R, et al. 2020. Task-induced brain connectivity promotes the detection of individual differences in brain-behavior relationships. NeuroImage. 207:116370.
- 43 Xiao L, Stephen JM, Wilson TW, Calhoun VD, Wang Y-P. 2019. Alternating diffusion map based fusion of multimodal brain connectivity networks for IQ prediction. *IEEE Trans Biomed Eng.* 66: 2140–2151.
- 44 Xiao L, Stephen JM, Wilson TW, Calhoun VD, Wang Y-P. 2020. A manifold regularized multi-task learning model for IQ prediction from two fMRI paradigms. IEEE Trans Biomed Eng. 67:796–806.
- 45 Keller AS, et al. 2023. Personalized functional brain network topography is associated with individual differences in youth cognition. Nat Commun. 14:8411.
- 46 Zhao W, et al. 2023. Task fMRI paradigms may capture more behaviorally relevant information than resting-state functional connectivity. *NeuroImage*. 270:119946.
- 47 Chen J, et al. 2022. Shared and unique brain network features predict cognitive, personality, and mental health scores in the ABCD study. Nat Commun. 13:2217.
- 48 Greene AS, Gao S, Scheinost D, Constable RT. 2018. Task-induced brain state manipulation improves prediction of individual traits. Nat Commun. 9:2807.
- 49 Mihalik A, et al. ABCD neurocognitive prediction challenge 2019: predicting individual fluid intelligence scores from structural MRI using probabilistic segmentation and kernel ridge regression in Adolescent Brain Cognitive Development Neurocognitive Prediction. In: Pohl KM, Thompson WK, Adeli E, Linguraru MG, editors. Lecture notes in computer science. Eds. Springer International Publishing, 2019. p. 133–142.
- 50 Makowski C, et al. 2024. Leveraging the adolescent brain cognitive development study to improve behavioral prediction from neuroimaging in smaller replication samples. *Cerebral Cortex*. 34: bhae223.

- 51 Noble S, Scheinost D, Constable RT. 2021. A guide to the measurement and interpretation of fMRI test-retest reliability. *Curr Opin Behav Sci.* 40:27–32.
- 52 Elliott ML, Knodt AR, Hariri AR. 2021. Striving toward translation: strategies for reliable fMRI measurement. Trends Cogn Sci. 25: 776–787.
- 53 Elliott ML, et al. 2019. General functional connectivity: shared features of resting-state and task fMRI drive reliable and heritable individual differences in functional brain networks. *NeuroImage*. 189:516–532.
- 54 Knodt AR, et al. 2023. Test-retest reliability and predictive utility of a macroscale principal functional connectivity gradient. Hum Brain Mapp. 44:6399–6417.
- 55 Avery EW, et al. 2020. Distributed patterns of functional connectivity predict working memory performance in novel healthy and memory-impaired individuals. J Cogn Neurosci. 32:241–255.
- 56 Jiang R, et al. 2020. Gender differences in connectome-based predictions of individualized intelligence quotient and sub-domain scores. Cerebral Cortex. 30:888–900.
- 57 Wu J, et al. 2022. Cross-cohort replicability and generalizability of connectivity-based psychometric prediction patterns. *NeuroImage*. 262:119569.
- 58 Zou H, Hastie T. 2005. Regularization and variable selection via the elastic net. J R Stat Soc Series B Stat Methodol. 67:301–320.
- 59 Cortes C, Vapnik V. 1995. Support-vector networks. Mach Learn. 20:273–297.
- 60 Breiman L. 2001. Random forests. Mach Learn. 45:5-32.
- 61 Chen T, Guestrin C. XGBoost: a scalable tree boosting system. Proceedings of the 22nd ACM SIGKDD international conference on knowledge discovery and data mining, KDD '16. Association for Computing Machinery, 2016. p. 785–794.
- 62 Barrett-Young A, *et al.* 2022. Associations between retinal nerve fiber layer and ganglion cell layer in middle age and cognition from childhood to adulthood. JAMA Ophthalmol. 140:262–268.
- 63 Salthouse TA. 2004. Localizing age-related individual differences in a hierarchical structure. *Intelligence*. 32:541–561.
- 64 Weintraub S, et al. 2014. The cognition battery of the NIH toolbox for assessment of neurological and behavioral function: validation in an adult sample. J Int Neuropsychol Soc. 20:567–578.
- 65 Weschler D. Wechsler adult intelligence scale—fourth edition. APA PsycTests, 2008. p. 1–3.
- 66 Ji JL, et al. 2019. Mapping the human brain's cortical-subcortical functional network organization. NeuroImage. 185:35–57.
- 67 Glasser MF, et al. 2016. The human connectome project's neuroimaging approach. Nat Neurosci. 19:1175–1187.
- 68 Behzadi Y, Restom K, Liau J, Liu TT. 2007. A component based noise correction method (CompCor) for BOLD and perfusion based fMRI. NeuroImage. 37:90–101.
- 69 Barch DM, et al. 2013. Function in the human connectome: task-fMRI and individual differences in behavior. NeuroImage. 80:169–189.
- 70 Glasser MF, et al. 2013. The minimal preprocessing pipelines for the Human Connectome Project. *NeuroImage*. 80:105–124.
- 71 Tejavibulya L, et al. 2022. Predicting the future of neuroimaging predictive models in mental health. *Mol Psychiatry*. 27:3129–3137.

- 72 Candia J, Tsang JS. 2019. eNetXplorer: an R package for the quantitative exploration of elastic net families for generalized linear models. BMC Bioinform. 20:189.
- 73 Apley DW, Zhu J. 2020. Visualizing the effects of predictor variables in black box supervised learning models. J R Stat Soc Series B Stat Methodol. 82:1059–1086.
- 74 Friedman JH, Popescu BE. 2008. Predictive learning via rule ensembles. Ann Appl Stat. 2:916–954.
- 75 Rosenberg MD, Finn ES. 2022. How to establish robust brain-behavior relationships without thousands of individuals. Nat Neurosci. 25:835–837.
- 76 Makowski C, Nichols TE, Dale AM. 2024. Quality over quantity: powering neuroimaging samples in psychiatry. Neuropsychopharmacology. 50:58–66. https://doi.org/10.1038/ s41386-024-01893-4
- 77 Poulton R, Guiney H, Ramrakha S, Moffitt TE. 2023. The Dunedin study after half a century: reflections on the past, and course for the future. J R Soc N Z. 53:446–465.
- 78 Fair DA, et al. 2007. A method for using blocked and event-related fMRI data to study "resting state" functional connectivity. NeuroImage. 35:396–405.
- 79 Gratton C, et al. 2018. Functional brain networks are dominated by stable group and individual factors, not cognitive or daily variation. Neuron. 98:439–452.e5.
- 80 Fischl B, et al. 2002. Whole brain segmentation: automated labeling of neuroanatomical structures in the human brain. Neuron. 33:341–355.
- 81 Destrieux C, Fischl B, Dale A, Halgren E. 2010. Automatic parcellation of human cortical gyri and sulci using standard anatomical nomenclature. *NeuroImage*. 53:1–15.
- 82 Wechsler D. Wechsler Intelligence Scale for Children—Revised. Psychological Corporation, 1974.
- 83 Ding DY, Li S, Narasimhan B, Tibshirani R. 2022. Cooperative learning for multiview analysis. Proc Natl Acad Sci U S A. 119: e2202113119.
- 84 Pedregosa F, et al. 2011. Scikit-learn: machine learning in Python. J Mach Learn Res. 12:2825–2830.
- 85 Drucker H, Burges CJ, Kaufman L, Smola A, Vapnik V. Support vector regression machines. In: Advances in neural information processing systems, Vol. 9. Morgan Kaufmann, San Mateo, 1996. p. 155–161.
- 86 Efron B, Tibshirani RJ. An introduction to the bootstrap. CRC Press, New York, 1994.
- 87 Vos De Wael R, et al. 2020. BrainSpace: a toolbox for the analysis of macroscale gradients in neuroimaging and connectomics datasets. Commun Biol. 3:103.
- 88 Abraham A, et al. 2014. Machine learning for neuroimaging with scikit-learn. Front Neuroinform. 8. doi:10.3389/fninf.2014.00014.
- 89 Shrout PE, Fleiss JL. 1979. Intraclass correlations: uses in assessing rater reliability. Psychol Bull. 86:420–428.
- 90 Cicchetti DV, Sparrow SA. 1981. Developing criteria for establishing interrater reliability of specific items: applications to assessment of adaptive behavior. Am J Ment Defic. 86:127–137.